Author's response to reviews

Title: The CTLA4 variants may interact with the IL23R- and NOD2-conferred risk in development of Crohn's disease

Authors:

Ondrej Hradsky (ondrej.hradsky@lfmotol.cuni.cz)
Petra Dusatкова (petra.dusatkova@seznam.cz)
Martin Lenicek (mleni@centrum.cz)
Jiri Bronsky (bronsky@email.cz)
Jiri Nevoral (jiri.nevoral@seznam.cz)
Libor Vitek (vitek@cesnet.cz)
Milan Lukas (milan.lukas@email.cz)
Ivana Zeniskova (i.zeniskova@volny.cz)
Ondrej Cinek (ondrej.cinek@lfmotol.cuni.cz)

Version: 4 Date: 5 June 2010

Author’s response to reviews:

1) Author list: the list of authors in the manuscript have been rewritten exactly as it is in the submission system.
2) Authors contributions: The sentence "All authors read and approved the final manuscript.” have been added to Authors' contributions part.
3) Tables: Tables have been renumbered in ascending numerical order.